Author's response to reviews

Title: Progesterone for the prevention of preterm birth in women with multiple pregnancies: the AMPHIA trial

Authors:
Arianne C Lim (A.C.Lim@amc.nl)
Kitty WM Bloemenkamp (k.w.m.bloemenkamp@lumc.nl)
Kees Boer (K.Boer@amc.nl)
Johannes J Duvekot (j.j.duvekot@erasmusmc.nl)
Jan Jaap HM Erwich (j.j.h.m.erwich@og.umcg.nl)
Tom HM Hasaart (tom.hasaart@cze.nl)
Pieter Hummel (p.hummel@mca.nl)
Ben WJ Mol (b.w.mol@amc.uva.nl)
Jos PM Offermans (jof@sgyn.azm.nl)
Charlotte M van Oirschot (c.v.oirschot@elisabeth.nl)
Job G Santema (j.santema@znb.nl)
Hubertina CJ Scheepers (L.Scheepers@obgyn.umcn.nl)
Willem A Schols (w.schols@meandermc.nl)
Frank PHA Vandenbussche (frank.vandenbussche@lumc.nl)
Maurice GAJ Wouters (mgai.wouters@vumc.nl)
Hein W Bruinse (h.w.bruinse@umcutrecht.nl)

Version: 3  Date: 11 June 2007

Author's response to reviews:

Editor-in-Chief
BMC Pregnancy and Childbirth

Amsterdam, 11 June 2007

Subject: Study Protocol "17-Alpha hydroxyprogesterone in Multiple pregnancies to Prevent Handicapped InfAnts: the AMPHIA trial [ISRCTN 40512715]"

Dear BMC Pregnancy and Childbirth,

I hereby resubmit the study protocol entitled "17-Alpha hydroxyprogesterone in Multiple pregnancies to Prevent Handicapped InfAnts: the AMPHIA trial [ISRCTN 40512715]" for online publication in BMC Pregnancy and Childbirth.

The following changes have been made to the manuscript:
- Affiliations - Each affiliation is now written in full and provided in the following format: department, institute, city, country.
- Author list - The list of authors in the manuscript is now written exactly as they are in the submission system.
- Running head/Running title has been removed.
- References - The reference list now contains all authors' names, if there are 30 or below. The term 'et al.' is no longer used.
- Author's contributions has been changed to Authors' contributions.

Yours sincerely,
Arianne Lim, MD, corresponding author

Academic Medical Centre Amsterdam
Department of Obstetrics and Gynaecology
H4-250
P.O.Box 22660
1100 DD Amsterdam
The Netherlands
Tel: +31 20 566 4256
Email: A.C.Lim@amc.nl